Author's response to reviews

Title: An Intra-tendonous ganglion cyst causing impingement between the anterior cruciate ligament and anterior root of the medial meniscus: a case report.

Authors:

mei guolong (meiguolong1@yahoo.cn)
Gao zhi (gauss529@163.com)
Hu yong (meiguolong1@yahoo.cn)

Version: 8 Date: 2 October 2013

Author's response to reviews: see over
Dear Editor,

Tanks for you attention, I think I have Modified according to your instructions as follow:

In the revised, we stress out this type of ganglion cyst can be caused by injury of the anterior horn of the meniscus and surgery limited on the meniscus can solve the problem in the early time. thereby neglecting, this report is the first or not in the the conclusion section

With thanks for your consideration,

Yours Mei guo-long